Author’s response to reviews

Title: Partial mastectomy using manual blunt dissection (MBD) in early breast cancer

Authors:

Shinichiro Kashiwagi (spqv9ke9@view.ocn.ne.jp)
Naoyoshi Onoda (nonoda@med.osaka-cu.ac.jp)
Yuka Asano (asnyk0325@yahoo.co.jp)
Kento Kurata (kent6124@ybb.ne.jp)
Tamami Morisaki (spitz4_5@yahoo.co.jp)
Satoru Noda (s-noda@med.osaka-cu.ac.jp)
Hidemi Kawajiri (hkawajiri75@yahoo.co.jp)
Tsutomu Takashima (tsutomu-@rd5.so-net.ne.jp)
Kosei Hirakawa (hirakawa@med.osaka-cu.ac.jp)

Version: 4 Date: 4 October 2015

Author’s response to reviews: see over
Editor-in-Chief
BMC Surgery

Dear Executive Editor

Dr. Tom Rowles

Re: MS: 1568770131701210 “Partial mastectomy using manual blunt dissection (MBD) in early breast cancer”

We greatly appreciate your invitation for us to re-submit our article. We would like to thank you for a number of comments and suggestions for improvement in our manuscript entitled “Partial mastectomy using manual blunt dissection (MBD) in early breast cancer” by Kashiwagi S et al.

I hope this revised manuscript can again be considered for publication in the BMC Surgery.

Sincerely,

Shinichiro Kashiwagi, MD
Department of Surgical Oncology, Osaka City University Graduate School of Medicine,
1-4-3 Asahi-machi, Abeno-ku, Osaka 545-8585, Japan
TEL; (+81) 6-6645-3838
FAX; (+81) 6-6646-6450
e-mail address; spqy9ke9@view.ocn.ne.jp
We have responded to the comments from Referee #1, as follows:

Referee #1

Dear Mark Trombetta

⇒ Thank you very much for the careful review of the reviewer.

Reviewer’s report:
The manuscript is fine now, with the exception of formatting of references [three authors only, followed by et al] Mark

⇒ We are so sorry for insufficient formatting of references. We revised formatting of references.
We have responded to the comments from Referee #2, as follows:

Referee #2
Dear Edward Chang

Reviewer’s report:
The revised version is improved and acceptable for publication after the necessary revisions were made to address the criticisms.
⇒ Thank you very much for the careful review of the reviewer.
We asked native check over the whole paper and proofread an English sentence.
(Forte Science Communications, Job No. R1507260)